Author's response to reviews

Title: Polymorphisms of XRCC1 genes and risk of nasopharyngeal carcinoma in the Cantonese population

Authors:

Yun Cao (yunsirsums@163.com)
Xiao-Ping Miao (Miaoxp@sina.com)
Ma-Yan Huang (hmy220@sina.com)
Ling Deng (dengling81@yahoo.com.cn)
Li-Fu Hu (Lifu.Hu@mtc.ki.se)
Ingemar Ernberg (Ingemar.Ernberg@mtc.ki.se)
Yi-Xin Zeng (yxzeng@gzsums.edu.cn)
Dong-Xin Lin (dlin@public.bta.net.cn)
Jian-Yong Shao (shjiany@mail.sysu.edu.cn)

Version: 5 Date: 9 May 2006

Author's response to reviews:

The BioMed Central Editorial Production Team:

We have changed the manuscript titled "Polymorphisms of XRCC1 genes and risk of nasopharyngeal carcinoma in the Cantonese population" according to the pending format requirements. Here we will re-submit the latest revision of this manuscript.

Thank you all for your kindly helps.

Best regards

Jian-Yong Shao, M.D., Ph.D, Professor
Dept. of Pathology
Sun Yat-Sen University Cancer Center
651 Dong Feng Road East
Guangzhou 510060
P.R. China
Email:shjiany@mail.sysu.edu.cn